Author’s response to reviews

Title: Association of trace metal elements with lipid profiles in type 2 diabetes mellitus patients. a cross sectional study

Authors:

Amare Wolide (amaju2002@yahoo.com)

Belay Zawdie (bellzolla2000@gmail.com)

Tilahun Alemayehu (tilalenmg2012@gmail.com)

Samuel Tadesse (sami2tadi@gmail.com)

Version: 2 Date: 08 Jul 2017

Author’s response to reviews:

Reviewer reports:

Khalid Siddiqui (Reviewer 1): Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format. Please overwrite this text when adding your comments to the authors. Include table in text make flows in English text after addition of comment.

Answer: We are not clear for the above comments. We are not received any comments except minor language edition in the manuscript. Concerning about the language edition we tried our best for the improvement and only two minor changes has been done in the abstract part. In the abstract method section line 1 "Jimma University specialized hospital" was changed as Jimma University Specialized Hospital. In the method section line 5 "In Pearson partial correlation " was changed to In Pearson, partial correlation.